Author's response to reviews

Title: Adiposity markers and risk of coronary heart disease in patients with type 2 diabetes mellitus

Authors:

Simone F Tonding (si_tonding@hotmail.com)
Flávia M Silva (flavia.moraes.silva@hotmail.com)
Juliana P Antonio (jupecanha.antonio@gmail.com)
Mirela J Azevedo (mirelajobimazevedo@gmail.com)
Luis H Canani (luishenriquecanani@gmail.com)
Jussara C Almeida (jcalmeida@hcpa.ufrgs.br)

Version: 2  Date: 8 June 2014

Author's response to reviews:

As suggested, we have corrected the format of mathematical formula for Conicity index (C-index) and Body Adiposity Index (BAI) in page 5.